Author's response to reviews

Title: Living with Myotonic Dystrophy; what can be learned from couples? A qualitative study

Authors:

Edith HC Cup (e.cup@reval.umcn.nl)
Astrid Kinébanian (A.Kinebanian@kpnmail.nl)
Ton Satink (Ton.Satink@han.nl)
Allan J Pieterse (A.Pieterse@reval.umcn.nl)
Henk T Hendricks (H.Hendricks@reval.umcn.nl)
Rob AB Oostendorp (R.Oostendorp@iq.umcn.nl)
Gert Jan van der Wilt (G.vanderWilt@ebh.umcn.nl)
Baziel GM van Engelen (B.vanEngelen@neuro.umcn.nl)

Version: 4 Date: 7 July 2011

Author's response to reviews:

Structure of the manuscript is correct now. This means that we have added a conclusion and abbreviations compared to the previous version

Highlighting and tracking have been removed

References completed

Manuscript checked carefully